Author's response to reviews

Title: Soft tissue perineurioma of the retroperitoneum in a 63-year-old man, CT and MR imaging findings: a case report

Authors:

Mayumi Yasumoto (mayumihoo@gmail.com)
Yoshiaki Katada (ktdmrad@tmd.ac.jp)
Reiko Matsumoto (rad@saitama-med.jrc.or.jp)
Akiko Adachi (akkia66@mn.em-net.ne.jp)
Koh Kaneko (byoribu@saitama-med.jrc.or.jp)

Version: 3 Date: 30 April 2010

Author's response to reviews:

We changed the title and restricted the abstract, as indicated by the Editor.
The ethnicity of the patient was also added: Japanese.
A conclusion section was included.
All newly added sections and changes are highlighted.
Several misspellings were corrected.

To the comments of Reviewer 1 (Andrea Mazzari),
The surgeons performed also splenectomy, as the tumor was first originated from
the pancreas, and possibility of malignancy could not be denied during the
operation.